\documentclass[12pt]{article}
% The above line tells the computer that this is an article.  You may also play around with other document styles (e.g. report), but article is a good starting point.
% The percent sign is a comment.  Characters following a percent sign are ignored by the compiler.

% You can call several packages before you start the document.
\usepackage[margin=1.25in]{geometry} % this packages has all sorts of features, the above is just an example...

\renewcommand{\baselinestretch}{1.25} %line spacing - set to 2 for doublespaced
\usepackage{graphicx} % use this for figures in your report.  
\usepackage{url} % read up on this yourself
\usepackage{fancyhdr} % read up on this yourself
%\usepackage[sort&compress]{natbib}
\usepackage[superscript]{cite}
\usepackage{amsmath, amsthm, amssymb}
\pagestyle{fancy}
\usepackage{float}
\usepackage{appendix}

% These are to generate sample text for this demo...you should not need them yourself.
\usepackage[english]{babel}
\usepackage{blindtext}

% This command starts the actual document...
\begin{document}


\title{Some Experiment Title\\
Lab Dates (E.g.: Sept. 9--26)}
\author{ Your Name
\\
Physics 3925 - Experimental Physics\\
Department of Physics and Astronomy \\
University of Lethbridge \\
\\
Lab Partner: Their Name\\
}
\date{\today}
\maketitle

\pagebreak

%% These will be useful to you later on...
%\begin{figure}[H]
%\begin{center}
%\includegraphics[width=0.5\textwidth]{}
%\caption{}
%\end{center}
%\label{}
%\end{figure}


%\begin{equation}
%\tag{}
%\end{equation}

%\begin{table}
%\caption{\label{tab:nickname} This is an exampel table caption.  Table captions go above the table.}
%\begin{tabular}{rcl}
%\hline
%Left column & Middle & Right \\
%\hline
%a & b & c \\
%z & y & x \\
%\hline
%\end{tabular}
%\end{table}


%\tableofcontents
%\setcounter{tocdepth}{2}

%\pagebreak
% figure out the difference betwqeen \section{} and \section*{}
\section*{Abstract}
\blindtext

\section{Introduction}
\label{sec:intro}

This is the introduction, and the first numbered section in your report.  You can refer to this section as Section \ref{sec:intro}, Sec.\ \ref{sec:intro}, or as \S \ref{sec:intro}; pick one style and stick with it.  You can also refer to later sections as follows: \S \ref{sec:theory}, \S \ref{sec:subsec}, \ref{sec:proc}, \S \ref{sec:AppA}, etc.

\Blindtext


\section{Theory}
\label{sec:theory}
\blindtext

\subsection{Some important other name}
\label{sec:subsec}
\blindtext

An equation is inserted as follows:
\begin{equation}
\label{eq:emc}
 E = mc^2 ,
\end{equation}
where $E$ is the energy, and \ldots.   Recall that equations are part of your sentence and need appropriate punctuation (i.e., end them with a period or comma typically).  You can refer to them later as Eq.\ \ref{eq:emc}, or in advance as in Eq.\ \ref{eq:mg}.  Another equation is:
\begin{equation}
\label{eq:mg}
F = m g + \alpha .
\end{equation}
Note how you can use greek letters in equations, and also in other math environments within the text (e.g., $\theta, \phi, \delta, \Delta$).  

\blindtext
\subsection{Another Subsection}
\label{sec:sub2}

You can have subsections as needed.

\blindtext



\section{Procedure}
\label{sec:proc}

\blindtext

\subsection{Equipment}
\label{sec:equip}

It is a good idea to have an equipment/aparatus section.  Here is a table, Tab.\ \ref{tab:nickname} (note that the table is somewhere else in the document...you will have to figure this out as you go...):

\begin{table}
\begin{center}
\caption{\label{tab:nickname} This is an example table caption.  Table captions go above the table.}
\begin{tabular}{rcl}
\hline
Left column title & Middle title & Right title \\
\hline
a & b & c \\
content & more content & yet more \\
\& separates rows & and lines are ended with & \ \ \\
z & y & x \\
\hline
\end{tabular}
\end{center}
\end{table}

\subsection{Setup}
\label{sec:setup}
\blindtext


\section{Results}
\label{sec:res}

This is a figure.  You can cite papers using the cite command, e.g., some references used said some really good things\cite{spiderman, superdave}.  You can cite other papers too\cite{paper1, halliday}.

\begin{figure}[H]
\begin{center}
\includegraphics[width=0.8\textwidth]{lsfig1}
\caption{\label{fig:fig1} Some figure caption.}
\end{center}
\end{figure}

An example list is as follows
\begin{itemize}
\item Something
\item can
\item go
\item here
\end{itemize}

And an example enumerated list is as follows
\begin{enumerate}
\item Something
\item can
\item go
\item here
\end{enumerate}

\blindtext

You can also have figures within sub-folders, as in Fig.\ \ref{fig:fig2} below (note the location of this file in the zipped folder...).

\begin{figure}[H]
\begin{center}
\includegraphics[width=0.8\textwidth]{lsfig2}
\caption{\label{fig:fig2} Some other figure caption.}
\end{center}
\end{figure}

\blindtext


Results tables are also good.  E.g., the following visible features were noted :

\begin{table}[h]
\begin{center}
\caption{\label{tab:tab2}  Some other results}
\begin{tabular}{|c|c|c|c|}
\hline
something & unique [units] & can [units] & go here [units] \\
\hline
blah & 5953.55        & 1.1732V       & 1.1732      \\ 
\hline
monkeys & 6762.03        & 1.3325        & 1.3325      \\ 
\hline
cows  & 1070.53        & 211         & 208.93    \\ 
\hline
\end{tabular}
\end{center}
\end{table}

\blindtext

\section{Discussion}
\label{sec:disc}

\Blindtext


\section{Conclusion}
\label{sec:conc}

\blindtext

\clearpage

\begin{thebibliography}{}
% you will need to follow up on how to do this properly.  You may also wish to use bibtex...
\bibitem{spiderman}"Article Title." Source , date. 
\url{http://www.google.com}.

\bibitem{superdave} "Radiation." Journal title, date. 
\url{http://bing.com}.

\bibitem{paper1} J. A. Uthor and A. N. Other, "Article Title," Journal Name., 1967, 39, 125.

\bibitem{halliday}Halliday, David, and Robert Resnick. "43-4 Compton Effect." Fundamentals of Physics. New York: Wiley, 1988. 982-85. Print.


\end{thebibliography}

\clearpage

\begin{appendices}

\section{An example Appendix}
\label{sec:AppA}

\blinditemize

\section{Another one}
\label{sec:AppB}

\blindenumerate

\end{appendices}

\end{document}